Author's response to reviews

Title: A Cross-Sectional Study of Well Water Arsenic and Child IQ in Maine Schoolchildren

Authors:

Gail A Wasserman (wassermg@nyspi.columbia.edu)
Xinhua Liu (xl26@columbia.edu)
Nancy J Lolacono (njl2@columbia.edu)
Jennie Kline (jkk3@columbia.edu)
Pam Factor-Litvak (prf1@columbia.edu)
Alexander van Geen (avangeen@ldeo.columbia.edu)
Jacob L Mey (mey@ldeo.columbia.edu)
Diane Levy (dl2015@columbia.edu)
Richard Abramson (rich2paula@gmail.com)
Amy Schwartz (amy.schwartz@unh.edu)
Joseph H Graziano (jg24@columbia.edu)

Version: 2 Date: 12 November 2013

Author's response to reviews:

As requested a Conclusions Section has been added to the paper.